Author's response to reviews

Title: Visual Neglect in Posterior Cortical Atrophy

Authors:

Katia Andrade (andrade-k@hotmail.com)
Dalila Samri (dalila.samri@gmail.com)
Marie Sarazin (marie.sarazin@psl.aphp.fr)
Leonardo C Souza (leocruzsouza@hotmail.com)
Laurent Cohen (laurent.cohen@psl.aphp.fr)
Michel T Schotten (michel.thiebaut@gmail.com)
Bruno Dubois (bruno.dubois@psl.aphp.fr)
Paolo Bartolomeo (paolo.bartolomeo@upmc.fr)

Version: 3 Date: 28 June 2010

Author's response to reviews: see over
Paris, 28 June 2010

Dear Editor,

Thank you for your response to our revised manuscript entitled “Visual Neglect in posterior cortical atrophy” (by Andrade, Samri, Sarazin, Cruz de Souza, Cohen, Thiebaut de Schotten, Dubois and Bartolomeo).

In the present version, we addressed the two minor reviewers’ concerns. Below is a point-by-point description of the changes made, which are highlighted in yellow. The abstract includes 248 words and the main text contains 2471 words. The paper also includes three tables, one figure and 29 references.

We look forward to hearing from you.

Many thanks,
Katia Andrade, MD, PhD student
Paolo Bartolomeo, MD, PhD